Author’s response to reviews

Title: Association of NOS3 gene polymorphisms with Essential Hypertension in Sudanese patients: a case control study

Authors:

Sahar Gamil (dr_sahar_sj@hotmail.com)
Jeanette Erdmann (jeanette.erdmann@iieg.uni-luebeck.de)
Ihab Abdalrahman (drtarawa1@hotmail.com)
Abdelrahim Mohamed (abdelrahim_osman@yahoo.com)

Version: 4 Date: 26 Oct 2017

Author’s response to reviews:

As stated in our previous email, there are two many indirect identifiers in the supplementary table, which contains the questionnaire data. We currently cannot publish this data with the number of identifiers, especially as there is no consent for publication from the participants. In order to anonymise this data, and to ensure that none of the participants can be identified, please remove the gender information and change the ages to ranges, e.g. 75 would be 70-75.

Response: done